If you worked with a partnerFamily Name:Family Name:Given Name:Given NameUserid:UserIdSection:Section:Section in which to hand assignment back:COSC 1030 — Assignment 2: Sparse ArraysDue: Wednesday, February 12, 1997 at 10:30 a.m.A professional report is expected with subparts for each task and item to be handed in, withappropriate documentation — not too much, not too little. We are looking for �the steak not thesizzle�. Use reasonable sized fonts and whitespace. Do not pack to meet the page limit.Programs are suitably structured in content and style, commented and tested.Teams of two are permitted. A team submits a joint report. Each author receives the samegrade.REPORT LIMIT 30 logical pages, includes everything you submit. Reportsthat are too long receive an F.Task GradeWeightWeighted ScoreOverall report_ _ _ _ _ _ *15% =_ _ _ _ _ _ _ Space report (3.1)_ _ _ _ _ _ * 10% =_ _ _ _ _ _ _ Loop invariants_ _ _ _ _ _ * 10% =_ _ _ _ _ _ _ Programming_ _ _ _ _ _ * 30% =_ _ _ _ _ _ _ Comments_ _ _ _ _ _ * 10% =_ _ _ _ _ _ _ Testing_ _ _ _ _ _ * 25% =_ _ _ _ _ _ _ Total_ _ _ _ _ _ _ Letter grade_ _ _ _ _ _ _